Author's response to reviews

Title: Knowledge of modifiable risk factors of heart disease among patients with acute myocardial infarction in Karachi, Pakistan: a cross sectional study

Authors:

Muhammad S Khan (muhammad.saleem@aku.edu)
Fahim H Jafary (fahim.jafary@aku.edu)
Tazeen H Jafar (tazeen.jafar@aku.edu)
Azhar M Faruqui (nicvdedo@khi.comsats.net.pk)
Syed I Rasool (ssrasool@hotmail.com)
Juanita Hatcher (juanita.hatcher@aku.edu)
Nish Chaturvedi (n.chaturvedi@imperial.ac.uk)

Version: 7 Date: 26 April 2006

Author's response to reviews: see over
Dear Sir/Madam,

Thanks for your valuable revision of the paper.

I have addressed the comments both by providing with a point-by-point response to them and by revising the manuscript.

Formatting changes requested
-----------------------------

Our editors strongly advise you revise the conclusion of the abstract to read: 'Our findings highlight the lack of good level of knowledge of modifiable risk factors for heart disease among subjects admitted with AMI in Pakistan. There is urgent need for aggressive and targeted educational strategies in the Pakistani population.'

Conclusion has been revised

3 2 1 'Our findings highlight the lack of good level of knowledge of modifiable risk factors for heart disease among subjects admitted with AMI in Pakistan. There is urgent need for aggressive and targeted educational strategies in the Pakistani population.

Sincerely,

Tazeen H. Jafar, MD MPH
Director, Clinical Epidemiology Unit
Assistant Professor
Medicine and Community Health Sciences
The Aga Khan University, Karachi, Pakistan
Phone: 92-21-4930051 ext 4812
FAX: 92-21-493-4294
Email: tazeen.jafar@aku.edu